Author’s response to reviews

Title: Geographic surveillance of community associated MRSA infections in children using electronic health record data

Authors:

Lilly Hsi-Chih Immergluck (limmergluck@msm.edu)
Traci Leong (tleong@emory.edu)
Kevin Matthews (yrp4@cdc.gov)
Khusdeep Malhotra (tuf81845@temple.edu)
Trisha Chan Parker (trisha.parker@choa.org)
Fatima Ali (f.imanali@gmail.com)
Robert C. Jerris (robert.jerris@choa.org)
George S. Rust (george.rust@med.fsu.edu)

Version: 1 Date: 03 Dec 2018

Author’s response to reviews:

.